**Author’s response to reviews**

**Title:** CHANGES IN PLASMA ALDOSTERONE LEVEL AFTER WEIGHT LOSS BY BARIATRIC SURGERY IN MORBIDLY OBESE PATIENTS

**Authors:**

preaw suwannasrisuk (praw110@gmail.com; preaws@nu.ac.th)

Patchaya Boonchaya-anant (b_patchaya@yahoo.com)

Natnicha Houngham (natnicha_houngham@hotmail.com)

Suthep Udomsawaengsup (Suthep.u@gmail.com)

Sarat Sunthornyothin (saratmd@gmail.com)

**Version:** 2  **Date:** 26 Mar 2019

**Author’s response to reviews:**

1. Please include an availability of data and materials statement in your declarations.

   (Availability of data and materials: data and materials are available upon request to the corresponding authors)

2. It has come to our attention that throughout your manuscript there is significant overlap with previously published works, in particular:

   https://dev.endocrine.org/meetings/endo-annual-meetings/abstract-details?ID=26788


   (The overlapping that are highlighted in red#1 is from the same project but it was published as an abstract of less than 400 words (poster presentation) at the Endocrine Society meeting. The full manuscript has never been published anywhere.)
The statement "This study was presented as a poster presentation at 2016 Endocrine Society meeting in Boston, USA." was added to the Disclosures in the manuscript.